Author's response to reviews

Title: Psychosocial and other working conditions in relation to body mass index in a representative sample of Australian workers

Authors:

Aleck S Ostry (ostry@interchange.ubc.ca)
Samia Radi (srad@primusonline.com.au)
Amber M Louie (ammlouie@interchange.ubc.ca)
Anthony D LaMontagne (alamonta@unimelb.edu.au)

Version: 8 Date: 29 December 2005

Author's response to reviews:

The changes requested in formatting have been made as follows:
1. Title and abstract are identical in submitted form and in manuscript.
2. Author list is identical in submitted form and in manuscript.
3. Title is in sentence case in both submitted form and manuscript.
4. Qualifications have been removed.